Author’s response to reviews

Title: Heterogeneity in high-risk prostate cancer treated with high-dose radiation therapy and androgen deprivation therapy

Authors:

Daniel Cagney (danielcagney@yahoo.com)
Mary Dunne (mary.dunne@slh.ie)
Carmel O'Shea (carmel.o'shea@slh.ie)
Marie Finn (Marie.Finn@slh.ie)
Emma Noone (Emma.Noone@slh.ie)
Martina Sheehan (Martina.Sheehan@slh.ie)
Lesley McDonagh (Lesley.McDonagh@slh.ie)
Lydia O'Sullivan (Lydia.OSullivan@slh.ie)
Pierre Thirion (pierre@thirion.ie)
John Armstrong (armstrongtravelling@gmail.com)

Version: 2 Date: 22 Jun 2017

Author’s response to reviews:

We have made the required edits by adding a Declarations section to the paper